

LJMU Research Online

Mugenzi, LMJ, Tekoh, TA, Ntadoun, ST, Chi, AD, Gadji, M, Menze, BD, Tchouakui, M, Irving, H, Wondji, MJ, Weedall, GD, Hearn, J and Wondji, CS

Association of a rapidly selected 4.3kb transposon-containing structural variation with a P450-based resistance to pyrethroids in the African malaria vector Anopheles funestus

http://researchonline.ljmu.ac.uk/id/eprint/24897/

Article

Citation (please note it is advisable to refer to the publisher's version if you intend to cite from this work)

Mugenzi, LMJ, Tekoh, TA, Ntadoun, ST, Chi, AD, Gadji, M, Menze, BD, Tchouakui, M, Irving, H, Wondji, MJ, Weedall, GD, Hearn, J and Wondji, CS (2024) Association of a rapidly selected 4.3kb transposon-containing structural variation with a P450-based resistance to pyrethroids in the

LJMU has developed LJMU Research Online for users to access the research output of the University more effectively. Copyright © and Moral Rights for the papers on this site are retained by the individual authors and/or other copyright owners. Users may download and/or print one copy of any article(s) in LJMU Research Online to facilitate their private study or for non-commercial research. You may not engage in further distribution of the material or use it for any profit-making activities or any commercial gain.

The version presented here may differ from the published version or from the version of the record. Please see the repository URL above for details on accessing the published version and note that access may require a subscription.

For more information please contact researchonline@ljmu.ac.uk

http://researchonline.ljmu.ac.uk/

http://researchonline.ljmu.ac.uk/



Citation: Mugenzi LMJ, Tekoh TA, Ntadoun ST, Chi AD, Gadji M, Menze BD, et al. (2024) Association of a rapidly selected 4.3kb transposon-containing structural variation with a P450-based resistance to pyrethroids in the African malaria vector *Anopheles funestus*. PLoS Genet 20(7): e1011344. https://doi. org/10.1371/journal.pgen.1011344

Editor: Giorgio Sirugo, University of Pennsylvania, UNITED STATES OF AMERICA

Received: July 30, 2023

Accepted: June 17, 2024

Published: July 29, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pgen.1011344

Copyright: © 2024 Mugenzi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All genomic data sets are available from the European Nucleotide Archive. Pooled template whole-genome RESEARCH ARTICLE

Association of a rapidly selected 4.3kb transposon-containing structural variation with a P450-based resistance to pyrethroids in the African malaria vector *Anopheles funestus*

Leon M. J. Mugenzi ¹*, Theofelix A. Tekoh ^{1,2}, Stevia T. Ntadoun¹, Achille D. Chi¹, Mahamat Gadji ^{1,3}, Benjamin D. Menze¹, Magellan Tchouakui¹, Helen Irving⁴, Murielle J. Wondji^{1,4}, Gareth D. Weedall^{4,5}, Jack Hearn ^{4,6}, Charles S. Wondji^{1,4}*

LSTM Research Unit, Centre for Research in Infectious Diseases (CRID), Yaoundé, Cameroon,
Department of Biochemistry and Molecular Biology, Faculty of Science University of Buea, Buea,
Cameroon, 3 Department of Microbiology, The University of Yaounde I, Yaounde, Cameroon, 4 Vector
Biology Department, Liverpool School of Tropical Medicine, Pembroke Place, Liverpool, United Kingdom,
School of Biological and Environmental Sciences, Liverpool John Moores University, Liverpool, United
Kingdom, 6 Centre for Epidemiology and Planetary Health, Department of Veterinary and Animal Science,
North Faculty, Scotland's Rural College, An Lòchran, 10 Inverness Campus, Inverness, Scotland, United

* leon.mugenzi@crid-cam.net (LMJM); charles.wondji@lstmed.ac.uk (CSW)

Abstract

Deciphering the evolutionary forces controlling insecticide resistance in malaria vectors remains a prerequisite to designing molecular tools to detect and assess resistance impact on control tools. Here, we demonstrate that a 4.3kb transposon-containing structural variation is associated with pyrethroid resistance in central/eastern African populations of the malaria vector Anopheles funestus. In this study, we analysed Pooled template sequencing data and direct sequencing to identify an insertion of 4.3kb containing a putative retro-transposon in the intergenic region of two P450s CYP6P5-CYP6P9b in mosquitoes of the malaria vector Anopheles funestus from Uganda. We then designed a PCR assay to track its spread temporally and regionally and decipher its role in insecticide resistance. The insertion originates in or near Uganda in East Africa, where it is fixed and has spread to high frequencies in the Central African nation of Cameroon but is still at low frequency in West Africa and absent in Southern Africa. A marked and rapid selection was observed with the 4.3kb-SV frequency increasing from 3% in 2014 to 98% in 2021 in Cameroon. A strong association was established between this SV and pyrethroid resistance in field populations and is reducing pyrethroid-only nets' efficacy. Genetic crosses and qRT-PCR revealed that this SV enhances the expression of CYP6P9a/b but not CYP6P5. Within this structural variant (SV), we identified putative binding sites for transcription factors associated with the regulation of detoxification genes. An inverse correlation was observed between the 4.3kb SV and malaria parasite infection, indicating that mosquitoes lacking the 4.3kb SV were more frequently infected compared to those possessing it. Our findings highlight the underexplored role and rapid spread of SVs in the evolution of insecticide resistance and provide additional tools for molecular surveillance of insecticide resistance.

sequencing data are available under study accessions PRJEB13485 (Malawi 2002 and Malawi 2014), PRJEB24384 (Ghana, Benin, Cameroon and Uganda) and PRJEB35040 (Mozambique 2002, 2016; DRC-Kinshasa and Mikalayi). The sequences of the 5.3kb intergenic region of CYP6P5 and CYP6P9b have been submitted to GenBank accession number: OR000399 and the nucleotide sequence of the 4.3kb SV is provided in supporting file S1 Text. The underlying numerical data for all graphs and summary statistics have been provided as Supporting Information (S6 Table).

Funding: This research was funded by the Bill and Melinda Gates Foundation (INV-006003 to CSW) and a Wellcome Trust Senior Research Fellowship grant to CSW (217188/Z/19/Z). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Anopheles funestus is an important malaria transmitting mosquito whose control relies on pyrethroid insecticides. The ability to control malaria transmitting mosquitoes is threatened by the emergence of insecticide resistance. Developing effective strategies to prevent, slow, or overcome resistance require an understanding of the underlying genetic mechanisms. In this study, we identified a transposon containing insertion in *Anopheles funestus* mosquitoes in Central/East Africa but absent in other regions. Tracking its distribution and frequency over the years revealed a strong selection with frequencies reaching fixation in less than 3 years. Phenotype/genotype association studies revealed a strong correlation between pyrethroid resistance and genetic variant genotype but negative correlation with plasmodium parasite infection in mosquitoes. These findings shed light on the often-over-looked role of structural variations in the evolution of insecticide resistance, offering valuable insights and tools for monitoring and addressing this critical issue in malaria control.

Introduction

The control of vector-borne diseases, which accounts for more than 17% of all infectious diseases reported in 2020, still greatly relies on using insecticides despite a recently recommended vaccine [1]. Insecticides have effectively reduced the number of cases and deaths due to these diseases, with over 80% of the reduction in malaria cases between 2000 and 2015 attributed to their use [2]. All LLINs currently used (PermaNet 2.0, Olyset, Duranet), including the new generation nets (PermaNet 3.0, Olyset plus and Interceptor G2), contain a pyrethroid insecticide due to their high potency and low toxicity in humans [3]. However, malaria vectors have developed resistance to pyrethroids which has spread widely in field populations [4], severely affecting our ability to control *Anopheles* with evidence that it could be impacting malaria transmission [3,5].

Insecticide toxicity can act as intense selective pressure, leading to the rapid evolution of resistance through the overexpression or modified activity of detoxification enzymes such as cytochrome P450s (Weedall et al., 2019), alteration of the target site [6], thickening of the cuticle [7] and behavioural changes [8]. The widespread insecticide resistance is a major global challenge threatening the efficacy of current and future vector control tools [9]. In the context of pyrethroid resistance, mutations primarily associated with its target, the voltage-gated sodium channel, are common. However, in *An. funestus*, no Kdr mutations have been identified [10]. Therefore, pyrethroid resistance in this species is predominantly metabolic in nature. Most studies on the molecular bases of insecticide resistance have focused on single nucleotide polymorphisms and small indels because they can be readily identified with short reads [11,12]. However, growing evidence shows that structural variants (SVs) contribute to adaptive mechanisms, including insecticide resistance [13].

SVs represent an essential source of genetic variation and are defined as large DNA sequence variations, including duplications, deletions, insertions, inversions, mobile-element transpositions, and translocations throughout the genome [14]. Structural variants are abundant across chromosomes. They are frequently found near genes where they are often associated with expression and likely contribute to phenotypic variations [15] and are predominantly shaped by transposons [16]. Decades of research have shown that the alteration of cis-regulatory regions by SVs can lead to perturbation of gene expression and phenotype [15,17]. In *Anopheles* mosquitoes, gene copy number variations have been identified and correlated with increased expression of insecticide resistance-associated genes [13,18]. A 6.5kb

structural variant in *An. funestus* was recently shown to be associated with the increased expression of two cytochrome P450 genes whose overexpression confers high resistance to pyrethroids [19] in Southern Africa and is absent elsewhere (West, Central and East Africa) suggesting a restriction to gene flow [20]. Population studies have shown a clear demarcation between Southern and Central/Western/Eastern populations with the clustering of Ghanaian, Cameroonian and Ugandan populations [20]. This points to the possibility of resistance-associated mutations not only emerging independently within a population but also spreading from another resistant population as seen for the cytochrome P450s *CYP6P9a* and *b* in southern Africa [21].

A previous study on the promoter region of an insecticide resistance gene *CYP6P9b* in *An*. *funestus* showed that this gene's 1kb upstream region failed to amplify in certain countries (Uganda and Cameroon) while it was successfully amplified in most regions [22]. Therefore, it was hypothesised that a structural variant in this region could prevent PCR amplification.

In the current study, we investigated whether a structural variant insertion in a cytochrome P450 locus can modify the pyrethroid resistance phenotype in *An. funetus*. We discovered a 4.3 kb insertion in the p450s loci rp1, previously identified by QTL mapping, housing two insecticide resistance genes, CYP6P9a and CYP6P9b [23]. This insertion was found across Central, East, and West Africa but was absent in southern regions. Monitoring in *An. funestus* populations of Cameroon revealed strong selection on this structural variant, suggesting it may be adaptive or linked to nearby adaptive mutations. Genetic crosses indicated a robust association between this variant and resistance to pyrethroids, as well as overexpression of nearby genes.

Materials and methods

1 Mosquito samples

This study involved field-collected mosquitoes and an insecticide-susceptible laboratory strain FANG. Mosquito collections were conducted in four villages in Cameroon, where An. funestus s.s. is the predominant vector. The selected villages and corresponding years of collection are as follows: Gounougou (9°03'00"N, 13°43'59"E) in 2017, 2020, and 2021; Elende (3° 41'57.27"N, 11°33'28.46"E) in 2020; Mibellon (6°46'N, 11°70'E) in 2018 and 2020; Tibati in 2021 (6°28' N, 12°37' E) and Elon 4°13'49"N, 11°36'04"E in 2021. Additionally, mosquitoes previously collected in these localities and others were utilized for temporal monitoring and African wide genotyping: Gounougou in 2014 (Menze et al., 2018); Mibellon in 2016 (Menze et al., 2018); Tibati in 2018 2018 [24]; Ghana in 2014 and 2021 [25,26]; Mozambique in 2015; Uganda-Tororo in 2014 [20] and Uganda-Mayuge in 2017; Kenya in 2018; Benin in 2022 and Tanzania in 2018. Indoor resting mosquitoes were collected with Prokopack aspirators between 06:00 am and 9:00 am, following verbal consent from the house owner. The number of blood-fed Anopheles females collected varied from one location to another with an average of 100 per day in Gounougou and Elende and 50 or less in some locations like Tibati. The collected blood-fed Anopheles females were kept for 5 days until fully gravid before putting them in 1.5ml tubes for forced egg-laying. The F_1 s were pooled and reared at the Centre for Research in Infectious Diseases (CRID) until the emergence of adults. F₁ mosquitoes from Gounougou (collected in 2018) were used for susceptibility testing and evaluating bed net efficacy.

Molecular identification followed the cocktail PCR method of Koekemoer et al. [27] to discriminate members of the *An. funestus* group and confirm the species as *An. funestus* s.s was done. The *An. funestus* s.s laboratory strain FANG, maintained at the insectary of CRID, was used for the crossing and qRT-PCR as the reference strains to determine gene expression.

2. Mosquito rearing

Insectaries were maintained under standard conditions at 26 ± 4 °C with a relative humidity of 70 ± 10%. Larvae from field and laboratory strains were fed with ground fish food (TetraMin tropical flakes, Tetra, Blacksburg, VA, USA), and adults were provided with 10% sucrose solution on cotton wool.

3. Establishing crossing between field and laboratory mosquitoes

The F_1S of field-caught mosquitoes from Elende and Mibellon were crossed with the FANG *An. funestus* colony to allow segregation of the resistance allele already at a very high frequency in the population. F_1 larvae from the field mosquitoes and larvae from the FANG colony were reared until the pupal stage, then individually placed in 15ml tubes with 2ml of water and stoppered with cotton wool. This was done to obtain virgin female mosquitoes for each strain. Female F_1s of field-collected mosquitoes were crossed with male FANG to generate a field x FANG line using about 500 individuals from each. A reciprocal crossing using about 500 female FANG and 750 F_1 males from the field was successfully established. A restricted cohort of approximately 100 F_1 individuals was derived from the initial cross, subsequently subjected to intercrossing to generate the F_2 generation. Approximately 200 F_2 individuals were obtained and further subjected to crossing and blood feeding to produce the F_3 generation, resulting in a population increase to approximately 600 individuals. This F3 population was utilized for the bioassays. The complete study involved the generation of three distinct lines.

4. WHO insecticide susceptibility tests

The resistance profile to public health insecticides was determined using the WHO susceptibility bioassays protocol [28]. About 100 female mosquitoes aged 2–5 days old were exposed to the insecticide(s) for either 30 minutes (for the crossing) or 1 hour (for the F₁ field population), then transferred to holding tubes and provided with 10% sucrose solution. Fifty mosquitoes exposed to non-impregnated papers were used as a control group. The initial knockdown effect was scored immediately after exposure time, while the mortality rates were scored 24 hours post-exposure. These bioassays were conducted at 26 ± 2 °C and $80 \pm 10\%$ relative humidity.

5. WHO cone assays

The bio-efficacy of several bed nets was evaluated using the WHO cone assay protocol [29]. Cohorts of 50 mosquitoes were exposed to 25cm x 25cm pieces of the netting of PermaNet 3.0 [Combination of 2.8 g/kg of deltamethrin on side panels and deltamethrin (4.0 g/kg) and PBO (25 g/kg) on roof, manufactured by Vestergaard Frandsen], PermaNet 2.0 ($5.5 \times 10-5$ kg/m² of deltamethrin, manufactured by Vestergaard Frandsen), Olyset ($8.6 \times 10-4$ kg/m² of permethrin, manufactured by Sumitomo Chemical), and Olyset Plus ($8.6 \times 10-4$ kg/m² of permethrin and $4.3 \times 10-4$ kg/m² of PBO, manufactured by Sumitomo Chemical). PermaNet 3.0 had 2 pieces from the roof and the side, as the top is impregnated with insecticide and PBO while the side has only the insecticide. Insecticide-free nets were used as controls to which mosquitoes were also exposed. Mosquitoes were exposed to the nets for 3 minutes in 5 replicates (each of 10 mosquitoes) and then transferred into paper cups. The knockdown effect was recorded at 1-hour post-exposure, and the mortality rate was scored 24 hours post-exposure.

6. Screening of the structural variant near *CYP6P5/CYP6P9b* from wholegenome sequences from different regions of Africa

SV was searched using PoolSeq data from An. funestus populations available under European Nucleotide bioprojects PRJEB24384, PRJEB13485, PRJEB35040, PRJEB35040, PRJEB24351 and PRJEB10294 [18, 20] for 12 populations of An. funestus sampled across Africa from 2002 to 2016. The locations and year of collection are Benin (Kpome, 2015); Ghana (Obuasi, 2014); Cameroon (Mibellon, 2015); Uganda (Tororo, 2014); Democratic Republic of Congo [Kinshasa (2015), Mikalayi (2015]; Malawi (Chikwawa, 2014); Mozambique [Palmeira (2016), Morrumbene (2002) and Zambia (Kaoma, 2013) plus the FUMOZ and FANG laboratory colonies. These samples underwent sequencing using an Illumina HiSeq 2500 platform (2×150) bp, paired-end). These were aligned to the An. funestus F3 genome [30] using BWA (v0.7.17-r1188) [31]. Reads were coordinate sorted, and duplicates were marked with Picard (v2.18.15) [32]. The subsequent alignments were visually inspected for evidence of an insertion in the intergenic region of CYP6P5 and CYP6P9b in IGV. In addition, non-reference transposon insertions were searched for computationally using TranSurVeyor (v1.0) [33]. The region with high coverage corresponding to the insertion sequence was submitted to RepBase, assigned the identifier TE-1_AFu, and subsequently annotated utilizing the CENSOR and InterProScan web services [34,35].

Visualising nucleotide diversity across the rp1 locus

Alignments were re-created with the *An. funestus* F3 genome (VectorBase release 57, Ghurye et al., 2019) for Tororo (Uganda), Mibellon (Cameroon) and Obuasi (Ghana) populations using BWA (v0.7.17-r1188) (Li et al., 2009). Reads were coordinate sorted, and duplicates were marked with picard (v2.27.2) [32]. Pileup files for each population were created using Samtools (v1.6) [31] with minimum phred-scaled base quality of 10 and mapping quality per read of 20. Population genetic statistics nucleotide diversity and Tajimas' D were generated for 1000 bp non-overlapping windows across chromosome 2 using npstat (v1.0) [36] for each population separately. Plots of nucleotide diversity from positions 8,525,000 to 8,575,000 incorporating the *rp1* resistance locus and 4.3 kb insertion site between *CYP6P5* and *CYP6P9b* were created using the karyoploteR package (v1.16.0) in R version 4.0.3 [37].

PCR amplification and sequencing of the entire *CYP6P5-CYP6P9b* intergenic region

To validate the poolseq results, which indicated the presence of a structural variant in the upstream region of *CYP6P9b*, the entire intergenic region between *CYP6P5* and *CYP6P9b* was amplified. Primers were 6P9dplF CCC CCA CAG GTG GTA ACT ATC TGA A located at 19bp before the stop codon of *CYP6P5* and the 6P9Ra/b TAC ACT GCC GAC ACT ACG AAG located at 35bp after the start codon of *CYP6P9b* and the Phusion high fidelity DNA polymerase (Thermo Scientific) was used for the amplification. The Phusion Taq PCR mix consisted of 3µl of 5x HF buffer, 0.12µl of 25mM dNTPs, 10mM forward and reverse primers, 0.15µl Phusion Taq, 9.71µl of deionised water and 1µl of DNA for 15µl reaction. Thermocycler conditions were: pre-denaturation at 98°C for 1 minute; 35 cycles of denaturation at 98°C for 10 seconds, annealing at 62°C for 30 seconds, extension at 72°C for 4 minutes; a final extension at 72°C for 10 minutes. PCR amplicons were stained with Midori Green Advance DNA Stain (Nippon Genetics Europe GmbH) and visualised and size-scored following (1%) agarose gel electrophoresis, using an ENDURO GDS (Labnet) UV transilluminator. PCR products were gel-purified using the QIAquick Gel Extraction Kit (Qiagen, Hilden, Germany), ligated into

PJET1.2 blunt-end vectors and Sanger-sequenced from each end of the 5.3 kb fragment using the plasmid-specific sequencing primers pJET1.2F and pJET1.2R. To sequence the complete fragment, four additional internal sequencing primers were used (S1 Table). Sanger sequencing was done at GENEWIZ; UK. Sequence data were analysed with BioEdit software [38].

Design of a simple PCR assay to detect the 4.3kb SV and analysis of its distribution and association with pyrethroid resistance

A PCR was designed to discriminate between mosquito samples with the 4.3kb structural variant and those without, consisting of 3 primers. Two primers (4.3kb_INSL_F: GGG GCG CTT TAG TTG AGA T and 4.3kb_INSR_R: CAC GTT TCA AGT GCA GGT GA) form a pair flanking the insertion but, due to the size of the insertion, amplify only for samples lacking the insertion, to produce a 281bp amplicon. A third primer (4.3kb INS R: CAT ACG CCT CTC CAG CAT GGA) binding within the structural variant forms a pair with 4.3kb_INSL_F to give a 780bp product only from samples containing the insertion. PCR amplification was done using the Kapa Taq PCR kit (Kapa Biosystems) with a 15µl reaction mix composed of 10x buffer A, 0.75µl of 25mM MgCl₂, 0.12µl of 25mM dNTPs, 0.5µM of each primer, 0.12µl of Kapa Taq enzyme, 10.49µl of deionised water and 1ul of genomic DNA. Thermocycler conditions were: pre-denaturation at 95°C for 5 minutes; 35 cycles of denaturation at 94°C for 30 seconds, annealing at 60°C for 30 seconds, extension at 72°C for 1 minute; a final extension at 72°C for 5 minutes. PCR products were visualized as described above. After optimisation, these assays were used to investigate any possible association between this structural variant and pyrethroid resistance using field F1 and laboratory mosquitoes by genotyping mosquitoes dead and alive after insecticide bioassays.

The spatio-temporal distribution of this structural variant in *An. funestus* s.s. populations collected in different location across Africa (Ghana, Cameroon, Kenya, Uganda, Tanzania, and Mozambique) was investigated. Genomic DNA samples from previous collections from across Africa [20] were also used.

Assessment of the expression of genes in proximity to the 4.3kb structural variant using qRT-PCR

To determine if the 4.3kb SV affected the expression of nearby genes, qPCR was used to compare the expression of 3 genes (CYP6P5, CYP6P9a, and CYP6P9b) found near it. The Elende x Fang cross was used to generate pools of mosquitoes with different genotypes: homozygous for SV (SV+/SV+), heterozygous for SV (SV+/SV-), and homozygous no SV (SV-/SV-). After rearing Elende x Fang mosquitoes to the F3 generation, adult females aged 3-5 days old were collected and kept at -80°C. DNA was extracted from the legs as described previously [19] and used for the 4.3kb insertion genotyping. Bodies were kept in RNAlater (ThermoFisher scientific) and stored at -80°C until genotyping was completed. The bodies were grouped in triplicates of 8 mosquitoes, each according to their genotypes: SV+/SV+, SV+/SV-, and SV-/SV-. RNA was extracted by genotype using the Arcturus PicoPure RNA Isolation Kit (Life Technologies), and cDNA was synthesised using the Superscript III (Invitrogen) as previously described [19]. The qRT-PCR amplification of CYP6P5, CYP6P9a, and CYP6P9b were assessed relative to the susceptible FANG strain on the Agilent MX3005 using the standard protocol from [39]. Primers are listed in S2 Table. Relative expression and fold-change of each target genes were calculated according to the $\Delta\Delta$ CT method incorporating PCR efficiency [40] after normalisation with the An. funestus housekeeping ribosomal protein S7 (RSP7) and actin 5C genes.

Identification of putative transcription factor binding sites present in 4.3kb SV

To check the possible role of this SV in gene regulation, we analysed the DNA sequence of the 4.3kb SV using the CiiiDER software [41]. CiiiDER uses the MATCH algorithm to predict the transcription factor binding sites in a query set of DNA sequences. First, the JASPAR CORE non-redundant vertebrate transcription factors [42] were used as the position frequency matrix PFM transcription factor model. Then we selected transcript factors like Aryl hydrocarbon receptor (Ahr), Muscle aponeurosis fibromatosis (Maf), previously implicated in xenobiotic response [43,44].

Plasmodium infection in relation to the 4.3kb structural variant

Genotyping results for Obout 2016 samples revealed the presence of the three genotypes for the 4.3kb SV. Also, previous data had shown that mosquitoes collected in Obout in May 2016 had high Plasmodium infection rates. The Plasmodium infection rate was 38.7% (72/ 186), 79.2% of which was P. falciparum, 12.5% ovale-vivax-malariae OVM + and 8.3% mix infection [45]. Screening for *Plasmodium* infection using TaqMan assay was done on 186 whole mosquito specimens from Obout. The real-time PCR MX 3005 (Agilent, Santa Clara, CA, USA) system was used for the amplification [46]. Briefly, 2 μ L of gDNA for each sample was used as a template in a 3-step program with a pre-denaturation at 95°C for 10 mins, followed by 40 cycles of 15 sec at 95°C and 1 min at 60°C. The primers Plas-F (5'-GCT TAG TTA CGA TTA ATA GGA GTA GCT TG-3') and Plas R (5'-GAA AAT CTA AGA ATT TCA CCT CTG ACA-3') were used together with two probes tagged with fluorophores: Falcip+(TCT GAA TAC GAA TGT C) labelled with FAM to detect Plasmodium falciparum, and OVM+ (CTG AAT ACA AAT GCC) labelled with HEX to detect *Plasmodium ovale*, Plasmodium vivax, and P. malariae. P. falciparum samples and a mix of P. ovale, P. vivax, and P. malariae were used as positive controls. A sub-set of positive samples was subjected to Nested PCR to confirm and discriminate the species detected by TaqMan based on the protocol of [47]. Only the P. falciparum positives and non-infected were used to investigate the Plasmodium infection rates in mosquitoes with different genotypes for the 4.3kb SV. In total 79 mosquitoes were analysed.

Data analysis

The percentage mortality for the bioassays and percentage genotypic and allelic frequencies for the dead and alive groups were computed in Excel 2016 spreadsheet (Microsoft). In addition, the odds ratio statistical test was calculated on the medcalc website (https://www.medcalc.org/calc/odds_ratio.php) to determine the strength of association between genotypes and the ability to survive insecticide exposure. The proportion of each genotype in the alive and dead individuals was determined and then used for the odds ratio calculations. The proportion of the alive individuals were considered as subjects with positive outcome while the dead were considered as subject with negative outcome. Fisher exact test was used to assess the significance level for the differences observed in the bioassay results, and results were considered significant at P value less than 0.05. The null hypothesis (H0) was that there is no statistically significant association between the observed phenotype are due to random chance or factors unrelated to the genotype being studied. Student's t-test was used to compare the means of data obtained for qPCR to determine the level of significance.

Results

1. Identification of 4.3kb structural variant in *Anopheles funestus* populations

Examining the PoolSeq IGV alignments uncovered a significant spike in coverage within the intergenic region in Uganda in 2014 compared to other populations (Fig 1A) corresponding to a 9 bp transposon insertion target duplication at positions 8,556,411–8,556,420 on Chromosome 2, marked by the sequence "CAAATGTACA." While there was faint visual evidence of the insertion in Cameroon, it was absent in other PoolSeq populations. TranSurVeyor confirmed the insertion in Uganda at the same positions but not in Cameroon or any other population. The inserted fragment is found in the reference strain FUMOZ at positions 24,765,013–24,760,696 on Chromosome 3, within the intron of an uncharacterized gene (AFUN019979). Due to the mixed-template PoolSeq approach, accurately determining the frequency of the structural variant (SV) in Uganda and Cameroon in 2014 was not feasible. However, it seems to have been of intermediate to high frequency in Uganda and very low frequency in Camero-



Fig 1. Identification and genotyping of a 4.3kb structural variant. A) Screenshot of Uganda individuals from IGV with a bump (circled) where the breakpoints (transposon insertion site) occurred in the Cyp6P5-Cyp6P9b intergenic region. Reads in red represent discordant reads. B) PCR amplification of the CYP6P9b and CYP6P5 intergenic region. Lane1 and 19 are the 1-kilobasepair DNA size marker;2 to 16 are field samples from Cameroon-Mibelong (2016); 17 is the negative control and 18 is a positive control. C) Schematic representation of CYP6P9b and CYP6P5 intergenic region with the structural variant and without the structural variant with the 2 open reading frames identified. D) Schematic representation of CYP6P9b and CYP6P5 intergenic region with the structural variant and site and without the structural variant Schematic representation of the 4.3kb SV diagnostic assay, consisting of 2 primers flanking the insertion site and 1 in the 4.3kb SV and electropherogram showing the different genotypes. +/+ = SV+/SV+, +/- = SV+/SV-.

https://doi.org/10.1371/journal.pgen.1011344.g001

oon. In the intergenic region, nucleotide diversity was lower in Uganda compared to Cameroon and FANG datasets, as well as other regions (S1 Fig). This reduction was accompanied by low Tajima's D values (D<0) in the rp1 locus (including the intergenic region between *CYP6P5* and *CYP6P9b* genes) with negative values observed in Uganda, Ghana and Cameroon compared to FANG (S3 Table). These negative Tajima's D values indicate positive selection or selective sweep across this locus in these populations. Multiple peaks of low diversity in Uganda across the rp1 locus, compared to other populations, suggest complex regional evolutionary patterns not confined to the 4.3 kb insertion. Inverting the ratio of nucleotide diversity between Uganda and other populations revealed a flat profile, indicating little reduced diversity at other rp1 loci versus Uganda (S2 Fig).

2. Amplification and sequencing of the 4.3kb structural variant insertion site in Uganda and Cameroon populations

The entire intergenic region between *CYP6P5* and *CYP6P9b* was amplified, cloned and sequenced to confirm the presence of this structural variant in Uganda and Cameroon *An. funestus* s.s. populations. PCR amplicons of approximately 1kb (no insertion) and 5.4kb (with the insertion) (Fig 1B) were cloned into pJET1.2/blunt cloning vectors and Sanger-sequenced. To sequence the full fragment, four additional sequencing primers were used (S1 Table). Sequence reads were aligned to reconstruct the full 5.4 kb region. To improve the sequence, pooled template WGS data were aligned to the draft 5.4 kb intergenic sequence flanked by the two coding genes (accession number: OR000399). Results revealed a 4311bp fragment inserted 494bp downstream of the translation stop codon of *CYP6P5* and 494bp upstream of the translation start codon of *CYP6P9b*.

Analysis of the entire 5.4 kb intergenic region revealed the presence of 2 open reading frames corresponding to a putative transposable element. BLASTp identified these open reading frames to match a gag-like protein from *Culex pipens* (AAB86424.1) with percentage identity of 29% and a reverse transcriptase-like protein from *Aedes aegypti* (AAA29354.1) with percentage identity of 39% (Fig 1C). The insertion sequence encodes a retrotransposon in the Jockey superfamily of long interspersed nuclear elements (LINEs). Jockey retrotransposons do not contain long terminal repeats (nonLTR) and are restricted to arthropods in their distribution [48]. The shorter of the two ORFs with 394 amino acids (aas) encodes a zinc finger domain, while the longer domain (906 aas) encodes endonuclease/exonuclease/phosphatase and reverse transcriptase domains in line with other Jockey elements [48]. A CENSOR search in the RepBase database revealed that a Jockey retrotransposon in *An. gambiae* (Ag-Jen-1) as the only similar sequence with a blast alignment score ("bit") of 3,997 and similarity of 0.67. There is a run of 15 Adenine bases in the 3' UTR of the insertion sequence consistent with a polyA tail.

3. Design of a genotyping assay for detection of the 4.3kb SV and delimitation of its geographical distribution

The availability of the 4.3kb structural variant sequence and its flanking region facilitated the design of a simple PCR for genotyping its presence or absence in laboratory and field samples. This assay consists of three primers, two flanking this SV and one within it (Fig 1D), as previously designed for a nearby 6.5kb SV between *CYP6P9a* and *b* genes (Mugenzi et al., 2020). The assay was initially tested on Uganda (Tororo, 2014) and Cameroon (Mibellon, 2021) field caught (F₀) samples containing the insertion and showed a band at 780bp. Further genotyping of Uganda (Tororo 2014 and Mayuge, 2017) field caught (F₀) mosquito samples confirmed its presence and at a high frequency approaching fixation of the 4.3kb SV in Uganda mosquitoes with 100% frequency in Tororo and 97.83% in Mayuge with no mosquito found homozygous for



Fig 2. Spatial and temporal distribution of the 4.3kb SV across Africa. (A) Geographical distribution of the 4.3kb SV in F0 *An. funestus* samples collected across Africa showing elevated frequencies in Cameroon and Uganda and absence in Ghana and Mozambique. (B), (C) and (D) shows the temporal monitoring of the 4.3kb SV genotypes in Cameroon in (B) Tibati 2018–2020; Mibellon 2016–2020 and (C) Gounougou 2014–2021. All revealing a increase in the frequency of the SV up to fixation.

https://doi.org/10.1371/journal.pgen.1011344.g002

the wild SV- allele (Fig 2A). Similar results were obtained for the Cameroon mosquito populations collected in Elon (2021), Mibellon (2021), Gounougou (2021) and Elende (2020), with SV +/SV+ genotypic frequencies of 88%, 100%, 97% and 89%, respectively (Fig 2A). Exploring its distribution in other localities across Africa revealed that it was absent from Ghana (West Africa) in 2014 and Mozambique (southern Africa) in 2015 and present at a very low frequency in East African Tanzania (5% SV+/SV-) in 2018 (Fig 2A). Genotyping of the *An. funestus* FUMOZ and FANG lab colonies showed that this structural variant was absent from those colonies.

4. Temporal monitoring of the 4.3kb SV allele frequency suggests it has evolved under strong positive selection

The high frequencies of this structural variant observed in Central (Cameroon) and Eastern Africa (Uganda) suggested that it is under a strong positive selection. To investigate this

possibility, we measured the frequencies of this SV in *An. funestus* s.s. populations of Cameroon collected in 2014, 2016, 2017, 2018, 2020 and 2021. Samples were available for 3 locations: Tibati and Mibellon, both from the Adamawa region and Gounougou (Northern region).

In Tibati, a temporal comparison of the frequency of the 4.3kb structural variant in samples collected in 2018 revealed a 76% SV+ allele frequency and complete fixation of SV+ in 2021 (S3A Fig). In addition, all three genotypes (SV+/SV+, SV+/SV- and SV-/SV-) were detected in 2018 at frequencies of 79%, 9% and 19%, respectively, while in 2021, only the SV+/SV+ genotype was detected (Fig 2B), indicating positive selection on the 4.3kb structural variant in this location.

A similar pattern was observed in Mibellon, with the structural variant present at a low allelic frequency (26%) in 2016, increasing to 98% in less than 2 years (2018) and then to 100% in 2020 (S3B Fig). In 2016, mosquitoes without the 4.3kb SV (SV-/SV) were more common (59%) than those with the structural variant in either the homozygous or heterozygous state (39% SV+/SV+; 11% SV+/SV-) (Fig 2C). By 2018 and 2020, the SV-/SV- homozygote was no longer detected, indicating that this 4.3kb structural variant was also driven to fixation under positive selection in this region. Genotyping of this SV in samples collected in Gounougou in 2014, 2017, 2020 and 2021 revealed a similar selection pattern to those observed in Tibati and Mibellon, with 3% in 2014, 31% SV+ in 2017, 63% in 2020 and 98% in 2021. The SV+/SV + homozygous genotype was at a very low frequency of 3% in 2014, reached a frequency of 26% in 2017 which doubled to 54% by 2020 in about 3 years and then almost got to fixation in 2021 (Fig 2D). The rapid increases in allele frequency of the 4.3kb SV in these 3 regions indicate strong selection acting on this structural variant (or a tightly linked allele) in these mosquito populations.

Analysis of recent samples collected in Ghana and Benin (West Africa) in 2022 showed that this insertion is now present in these localities. In Ghana, this SV was identified at a frequency of 5% with only 3 heterozygotes (S4B and S4C Fig), while in the Benin 2022 samples, a higher frequency of 33% was observed (SV+). The genotypes were present at frequencies of 16% (SV +/SV+), 34% (SV+/SV-) and 50% (SV-/SV-) (S4D and S4E Fig).

5. The 4.3kb SV is associated with pyrethroid resistance

We next aimed to establish if this 4.3kb contributes to pyrethroid resistance, thus explaining its swift increase in frequency in field populations of *An. funestus* from Central and East Africa. To investigate this, we employed the following experimental approaches:

- 1. Genotyping of F₁ Gounougou alive and dead field samples from WHO tube bioassays.
- 2. Genotyping of F₁ Gounougou alive and dead field samples exposed to LLINs from cone assays.
- 3. Genotyping of mosquito progenies issued from field and laboratory crosses exposed to pyrethroids (deltamethrin, permethrin, and alpha-cypermethrin).

Across all experimental approaches, the presence of the 4.3kb SV was consistently associated with increased survival following exposure to various pyrethroids (deltamethrin, permethrin, and alpha-cypermethrin) (Fig 3). The strength of this association was evaluated using odds ratios (ORs) and chi-square tests (Tables 1 and 2). In general, mosquitoes homozygous for the SV (SV+/SV+) exhibited the highest likelihood of survival compared to heterozygotes (SV+/SV-) and those lacking the SV (SV-/SV-). Heterozygotes also demonstrated an enhanced ability to withstand pyrethroid exposure relative to SV-/SV- individuals. The survival probability increased with the number of SV+ copies a mosquito possessed, suggesting an additive



Fig 3. Association of 4.3kb SV with pyrethroid resistance. (A) Genotyping results of the 4.3kb SV among the Gounougou 2018 F1s alive and dead after deltamethrin exposure reveal a strong association between the 4.3kb SV and ability to survive. (B) Genotype frequencies of the 4.3kb SV among the Gounougou 2018 F1s alive and dead mosquitoes exposed to PermaNet2 .0 bed nets showing a positive association between the 4.3kb SV and resistance. (C) illustrate the strong association between 4.3kb SV and the ability to survive exposure to deltamethrin by looking at its frequency among dead and alive F3 Elende-Fang crossing mosquitoes. (D) and (E) illustrate the strong association between 4.3kb SV and the ability to survive exposure to insecticide exposure by looking at its genotype distribution among dead and alive F3 Mibellon-Fang crossing mosquitoes exposed to permethrin (D) and α -cypermethrin (E).

https://doi.org/10.1371/journal.pgen.1011344.g003

effect of the structural variant alleles. At the allelic level, the SV+ allele was significantly associated with survival following pyrethroid exposure, with odds ratios ranging from 1.8 to 116.4 across different experiments and pyrethroids tested. These consistent findings from multiple independent experiments provide strong evidence that the 4.3kb structural variant is associated with resistance to pyrethroids in *An. funestus* populations, likely contributing to the reduced efficacy of pyrethroid-based insecticide-treated bed nets [Detailed results for each experimental approach can be found in the Supplementary Materials S1 Text and S6 Table].

6. Impact of 4.3kb structural variant insertion on the expression of nearby genes

To assess potential effect of this structural variant on the expression of nearby genes (*CYP6P5*, *CYP6P9a*, and *CYP6P9b*), crosses between field individuals (Elende, fully homozygous for the

Insecticide/Bednet	Comparison	OR ^a	P value	CI ^b
	WH	O BIOASSAY		
Deltamethrin	SV+/SV+ vs SV-/SV-	69.8	0.0001	8.4 to 578.5
	SV+/SV- vs SV-/SV-	6.9	0.1380	0.5 to 87.3
	SV+/SV+ vs SV+/SV-	10.2	0.008	1.8-56.5
	SV+ vs SV-	8.4	< 0.0001	8.4-41.7
	WH	O Cone assays		
PermaNet 2.0	SV+/SV+ vs SV-/SV-	201	0.0002	12.15-3325.23
	SV+/SV- vs SV-/SV-	201	0.0002	12.15-3325.23
	SV+/SV+ vs SV+/SV-	1	1	0.02-50.89
	SV+ vs SV-	82.93	0.0021	4.98-1379.74
Olyset plus	SV+/SV+ vs SV+/SV-	4.5238	0.3493	0.2 to 106.7
	SV+/SV+ vs SV-/SV-	1.8	0.04	1.1 to 3.2
	SV+/SV- vs SV-/SV-	0.3818	0.5562	0.01 to 9.4
	SV+ vs SV-	1.8	0.04	1.0 to 3.3

Table 1. Association between insecticide susceptibility as determined by WHO tube bioassay or WHO cone bioassay and 4.3kb SV genotype in wild-caught, female Anopheles funestus from Gounougou Cameroon in 2018.

https://doi.org/10.1371/journal.pgen.1011344.t001

4.3kb SV) and the fully susceptible FANG lab strain (4.3kb SV completely absent) were intercrossed to the F_3 generation. Quantitative real-time PCR performed on pools of each of the 3 genotypes (SV+/SV+, SV+/SV- and SV-/SV-) relative to FANG revealed moderate expression of nearby genes with increased expression of *CYP6P9a* (downstream) and *CYP6P9b* (immediately downstream) but not of *CYP6P5* (upstream) in the SV+/SV+ pool only (Fig 4A). *CYP6P9a* was most expressed in SV+/SV+ with fold change (FC) of 18.7 (*P-value* = 0.008), while SV+/SV- and SV-/SV- genotypes showed no differential expression. Similarly, *CYP6P9b*'s expression was higher in the SV+/SV+ genotype (FC = 16.0, *P value* = 0.002) and low in the SV+/SV- and SV-/SV- genotypes. This indicates that possessing 2 copies of this

Table 2. Association between insecticide susceptibility as determined by WHO tube bioassay and 4.3kb SV genotype in Anophele	es funestus crossing between field
and FANG lab colony.	

Insecticide	Comparison	OR ^a	P value	CI ^b
	Elende	X Fang crossing		
Deltamethrin	SV+/SV+ vs SV-/SV-	1517	0.0003	28.6-80383.1
	SV+/SV- vs SV-/SV-	41	0.01	2.2-761.8
	SV+/SV+ vs SV+/SV-	37	0.016	2.0-689.9
	SV+ vs SV-	19.4	0.0001	9.5-39.7
	Mibelong	X Fang crossing		
Permethrin	SV+/SV+ vs SV-/SV-	17.71	0.0001	4.3-72.3
	SV+/SV- vs SV-/SV-	2.75	0.0135	0.7-10.4
	SV+/SV+ vs SV+/SV-	6.44	0.0018	2.0-20.8
	SV+ vs SV-	5.63	0.0001	3.1-10.4
alpha-cypermethrin	SV+/SV+ vs SV-/SV-	72.72	< 0.0001	18.9-278.8
	SV+/SV- vs SV-/SV-	24	< 0.0001	10.5-54.2
	SV+/SV+ vs SV+/SV-	3.05	0.009	0.8-11.2
	SV+ vs SV-	116.43	0.0001	5.5-24.5

^a Odds ratios

^b confidence interval.

https://doi.org/10.1371/journal.pgen.1011344.t002



Fig 4. impact of 4.3kb structural variant on expression of nearby genes and Plasmodium infection. (A) Differential qRT-PCR expression for different structural variant genotypes for three cytochrome P450 genes in the immediate vicinity of the 4.3kb SV. Error bars represent standard deviation (n = 3). ns = not statistically significant; * = significantly different at p < 0.05. (B) 4.3kb SV genotypes distribution among *plasmodium* infected and non-infected F0 mosquitoes collected from Obout-Cameroon 2016 showing that samples with the 4.3kb SV are less infected than those without the 4.3kb SV.

https://doi.org/10.1371/journal.pgen.1011344.g004

4.3kb SV enhances the expression of these genes. For *CYP6P5*, there was no difference in the expression level for the different genotypes. Screening for transcription factors known to regulate detoxification genes in the 4.3kb SV using CiiiDER software identified Ahr, ARNT and MAF binding sites. For Ahr/Arnt, 4 binding sites while for 17 MAF binding sites were determined with 3 for MAFG, 2 for MAFF and 9 for MAFb (S4 Table).

7. Assessing the association between the 4.3kb structural variant and *Plasmodium* infection

Genomic DNA samples extracted from whole mosquitoes collected from Obout in 2016 were screened to detect *Plasmodium* infection and grouped into 'infected' (n = 37) and 'non-infected' (n = 42). These were genotyped for the 4.3kb structural variant, showing a significant association between genotypes (SV+/SV+, SV+/SV- and SV-/SV-) and infection status (χ 2 = 7.0; P = 0.031). The *Plasmodium*-infected group comprised 16.2% SV+/SV+, 43.2% SV +/SV- and 40.5%, SV-/SV-showing that the homozygous SV+/SV+ were less infected than the SV+/SV- and SV-/SV- genotypes (Fig 4B). Hence the SV+/SV+ were less likely to be infected than SV-/SV- (OR: 4.05; CI: 2.2–7.3; P <0.001) while no significant difference was observed between SV+/SV+ and SV+/SV- genotypes (OR: 1; CI: 1.5–474.6; P = 0.05) (Tables 3 and S6). Comparison of the allelic distribution of the 4.3kb SV among infected and non-infected further supported the reduced *Plasmodium*-infected mosquitoes, with 57.8% SV+ being non-infected against 45.2% SV-. Most Plasmodium-infected mosquitoes had SV- alleles without the structural variant (62.1%), while only 37.8% had SV+ alleles (S4 Table). Hence the likelihood of mosquitoes being infected by plasmodium parasite decreases with copies of the SV (Table 3).

Comparison	OR ^a	P value	CI ^b
SV+/SV+ vs SV-/SV-	4.1	0.0017	1.7-9.9
SV+/SV- vs SV-/SV-	4.2	0.0001	2.1-8.68
SV+/SV+ vs SV+/SV-	1	0.99	0.47-2.1
SV+ vs SV-	2	0.02	1.13-3.51

Table 3. Associations between 4.3kb SV genotypes and likelihood of not being infected by Plasmodium parasite.

^a Odds ratios, ^b confidence interval.

https://doi.org/10.1371/journal.pgen.1011344.t003

Discussion

This study assessed transposon-based structural variations' role in cytochrome P450-mediated metabolic resistance to insecticides in malaria mosquitoes by characterising and studying the effect of a 4.3kb intergenic insertion in a P450 cluster previously associated with resistance in the malaria vector *An. funestus*. A 4.3kb structural variant insertion first detected in East Africa in 2014 was shown to have spread rapidly in Central Africa, notably throughout Cameroon and continuing to move westward on the continent. Analysis of patterns and selection speed of this locus showed its association with strong resistance to pyrethroids, reduced bed net efficacy and *Plasmodium* infection, providing evidence that transposon-based resistance mechanism could be an important driver of metabolic resistance in malaria vectors.

Rapid selection of transposon-based resistance in malaria vectors with 4.3kb fixation in less than 5 years

PoolSeq data analysis identified this 4.3kb insertion between the CYP6P5 and CYP6P9b loci on Chromosome 2, found only in Uganda and Cameroon samples of 2014 out of 8 countries assessed. This 4.3bk SV contains 2 open reading frames (a gag-like and reverse transcriptaselike proteins), which correspond to 2 (gag and pol) out of the 3 open reading frames that characterised LTR- retrotransposons. The gag encodes capsid proteins, pol encodes enzymes regulating the transposition of a mobile element, while the missing env encodes a product responsible for the recognition of cell receptors and the penetration of a virus into a cell [49]. The Drosophila melanogaster gag-related gene (gagr), a homolog to the Gypsy group of LTR retroelements, is possibly associated with the origin of new functions and the involvement in stress response in Drosophila species [50]. This SV was at a high frequency in Uganda and observed at a low frequency in Cameroon in 2014. Therefore, we hypothesized that this SV spread from Uganda to Cameroon as the likelihood of a retrotransposon inserting twice at precisely the same location appears to be relatively low. Genotyping of recently collected samples for the 4.3kb SV revealed its presence in West Africa in Ghana and Benin at low frequencies, suggesting that this resistance allele is migrating westward. Interestingly population structure analyses using ddRADseq, Poolseq, and microsatellites have revealed a low level of divergence between Cameroon and Ugandan populations of An. funestus indicating that there is likely little barrier to gene flow and increased introgression of alleles between them [20,21]. Temporal analysis of the changes in the allelic frequency of the 4.3kb SV in Cameroon collected across the years (2014–2021) revealed a rapid selection of this marker, with its frequency reaching fixation in less than 5 years. Such rapid selection indicates that this insertion or a nearby mutation will likely provide mosquitoes with an essential adaptive advantage. This is supported by the low nucleotide diversity and negative Tajima's D values in the CYP6P5 to CYP6P9b intergenic region (S1 Fig and S2 Table).

The high frequencies observed exhibit a pattern akin to the 6.5 kb structural variant previously identified in *An. funestus* populations from southern Africa, specifically in Malawi and Mozambique [19]. This variant was correlated with an observed increase in resistance to deltamethrin and permethrin in field populations. This 6.5kb SV was shown to increase in frequency from 5% in 2002 to about 90% in 2016 in Mozambique samples [19]. Similarly, in *An. gambiae*, a partial Zanzibar-like transposable element (TE) was identified upstream of *CYP6aa1* in association with two other mutations (nonsynonymous point mutation in *CYP6P4* (I236M) and a duplication of *the CYP6AA1* gene) in Uganda populations at high frequency and shown to have spread to Kenya, the Democratic Republic of Congo and Tanzania [51].

4.3kb structural variant is associated with pyrethroid resistance

This structural variant was shown to contain two putative open reading frames. Several transposable elements, including the well-characterized Accord insertion upstream of the *CYP6G1* in *Drosophila melanogaster* that confers resistance to DDT, have been identified in close proximity to metabolic resistance genes [52]. This insertion has spread worldwide in this species [53]. A strong association was found between presence of the 4.3 kb SV and the survival to pyrethroid exposure. Exposing field-collected samples from Gounougou and genetic crosses generated in the insectary, revealed that most of the survivors were homozygous SV+/SV + while heterozygous SV+/SV - and homozygous SV-/SV- were mostly dead. In addition, mosquitoes with the SV+ allele were more likely to survive bed net exposure using cone assays to the standard pyrethroids-only nets (PermaNet 2.0), while PBO-containing bed nets are more effective against these mosquitoes. The results are similar to the association found with the 6.5kb SV was 242.4 (OR) while for the 4.3kb SV it was 5.63 (OR). Further studies are needed in semi-field conditions using experimental huts as previously done for the other metabolic markers [18,19] to better understand the impact of this SV on bed net efficacy.

A Transposable element containing SV is associated with increased expression of nearby P450s genes and Pyrethroid resistance

Transposable elements are over-represented near P450s clusters in insects [52]. They can affect the expression of the gene(s) by bringing regulatory elements that can either repress or enhance the expression of nearby genes [54]. Genetic crosses were used to compare the expression of three genes located near this 4.3kb SV between the 3 genotypes SV+/SV+, SV+/SVand SV-/SV- as previously done for the 6.5kb SV [19]. Increased expression of 2 genes (CYP6P9a and CYP6P9b) was found only in the homozygote SV+/SV+. This higher expression of nearby P450s in the homozygote SV+/SV+ suggests that the transposon contained in this 4.3kb region may act as enhancers. CYP6P9a is found at about 4kb downstream of the 4.3kb structural variant. Increased expression was also seen in wildtype RNASeq for Ugandan versus susceptible FANG mosquitoes sampled in 2014 [18] but was less clear cut than the crossing data due to complex expression of CYP6P9b across Africa. The RNA sequencing data also unveiled additional P450s exhibiting higher expression levels compared to the ones under our investigation. Therefore, there is a potential that this insertion may be contributing to the increased expression of other genes located at a distance. In D. melanogaster, the up-regulation of CYP6G1 conferring resistance to a variety of insecticide classes [55] correlates with the presence of an Accord retrotransposon in the 5' UTR region and this retro-transposable element contains regulatory sequences capable of increasing the expression of CYP6G1 in detoxification organs [56]. The 4.3kb SV identified here could regulate the expression of CYP6P9a/b by

providing regulatory elements. However, further studies are needed to functionally validate this hypothesis using luciferase promoter assay or promoter assays in transgenic mosquitoes. Additionally, the findings indicate a fully recessive P450 expression phenotype, characterized by overexpression exclusively in SV+/SV+ individuals. The recessive impact of this SV on expression could be due to transvection phenomenon as previously reported in fruit flies [57]. When in heterozygosis (SV+/SV-), a strong repressor in the SV- allele might dilute the activating effect of the 4.3kb SV. For the resistance phenotype, correlation with the presence of the SV+ allele both in the heterozygote and homozygote individuals was observed. Hence the survival of SV+/SV- genotype does not correlate with increased expression of *CYP6P9a/b* genes indicating that other genes might be complementing the observed resistance phenotype. Further investigations, involving bioassays with transgenic mosquitoes overexpressing *CYP6P9a/b* alleles with and without the 4.3 kb SV could help to better understand its contribution to resistance.

Reduced Plasmodium infection in mosquitoes with 4.3kb SV

A negative association was found between the 4.3kb SV and the malaria parasite infection, with mosquitoes lacking the 4.3kb SV being more commonly infected than those with the 4.3kb SV. We might speculate that this is linked to 4.3kb SV-driven overexpression of CYP6P9a/b since P450s are known to produce reactive oxygen species [58] which can modulate mosquito immunity to *plasmodium* helping to eliminate the parasite [59]. The role of P450 cytochromes during Plasmodium invasion remains poorly understood. Previous studies have shown specific cytochrome P450s implicated in insecticide resistance are also differentially expressed during malaria parasite invasion in mosquitoes [60]. For example, CYP6M2 and *CYP6Z1* both highly expressed in pyrethroid resistant *An. gambiae* mosquitoes [61,62] were also highly overexpressed in response to *Plasmodium* infection [60], suggesting an interaction between insecticide detoxification and infection. The results obtained are the opposite of what had been previously obtained with the GSTe2 119F resistant markers, where an association was observed with the homozygote 119F/F genotype being more associated with the Plasmodium infection at the oocyst and sporozoite stage [63]. These preliminary results need further validation using experimental infection studies as previously done with the kdr resistance mutations [64] to decipher the impact of P450-based resistance on malaria transmission.

Conclusion

This study identified a 4.3kb transposon-containing structural variant on chromosome 2 within a cluster of cytochrome P450 genes in Ugandan and Cameroonian populations of *Anopheles funestus* and showed it to be: associated with pyrethroid resistance, associated with enhanced expression of nearby P450 genes; rapidly evolving under strong selection in Cameroon and is spreading westward in Africa. This study shows how genetic variation, such as transposable elements are linked to adaptive changes and rapidly selected in the mosquitoes carrying them. The molecular assay designed here will facilitate the detection and tracking of the spread of this transposon-based resistance and help assess its impact on control intervention and malaria transmission.

Supporting information

S1 Fig. Ratios of PoolSeq nucleotide diversity for 1kb windows across the *rp1* **locus between two populations closely related to Uganda, Cameroon and Ghana, and the susceptible lab strain FANG.** A peak indicates a loss of diversity in Uganda versus the comparator population. The SV insertion position is labelled with the red-dashed line and gene positions

are indicated below the plots, genes are labelled by their cytochrome P450s name or there VectorBase identifier if not a P450 gene. An upper-bound of 100 was placed on peaks to ensure interpretable plots, these represent windows with very little to zero diversity in Uganda. (TIFF)

S2 Fig. Nucleotide diversity (pi) ratios for 1kb windows across the rp1 locus of *An. funestus* between Uganda and Ghana, Cameroon, and FANG PoolSeq datasets. This is the reverse of S1 Fig in the main text as Ugandan diversity is divided by comparator populations. It shows no peaks indicative of a drop of nucleotide diversity in the denominator population unlike the peaks shown for Uganda in S1 Fig. (TIFF)

(IIFF)

S3 Fig. (A) Allelic frequencies of 4.3kb SV in Tibati showing a decrease in SV- allele and increase in SV+ over the time. (B) Allelic frequencies of 4.3kb SV in Mibellon showing a decrease in SV- allele and increase in SV+ over the time. (TIFF)

S4 Fig. Changes in frequencies of the 4.3kb SV over time in (A) Gounougou, (B and C) (D and E) Benin and Ghana. (TIFF)

S5 Fig. Association of 4.3kb SV with pyrethroid resistance. (A) Genotyping results of the 4.3kb SV among the Gounougou 2018 alive and dead deltamethrin post exposure reveal a strong association between the 4.3kb SV and ability to survive. B) illustrate the strong association between 4.3kb SV and the ability to survive exposure to deltamethrin by looking at its genotypic and allelic distribution among dead and alive F3 Elende-Fang crossing mosquitoes. C) illustrate the strong association between 4.3kb SV and the ability to survive exposure to Permethrin by looking at its genotypic and allelic distribution among dead and alive F3 mibellon-Fang crossing mosquitoes. D) illustrate the strong association between 4.3kb SV and the ability to survive exposure to α -cypermethrin by looking at its genotypic and allelic distribution among dead and alive F3 mibellon-Fang crossing mosquitoes. (TIFF)

S1 Text. SI results.

(DOCX)

S1 Table. Additional sequencing primers were used to sequence the 5.5 kb intergenic sequence.

(DOCX)

S2 Table. qRT-PCR Primers. (DOCX)

S3 Table. Tajima's D values in the intergenic region between CYP6P5 and CYP6P9b genes. (DOCX)

S4 Table. Putative transcription factors binding sites in 4.3kb structural variants generate by CiiiDER software.

(DOCX)

S5 Table. Allelic frequencies of the 4.3kb SV among plasmodium infected and non-infected mosquitoes. (DOCX) S6 Table. Genotyping results of 4.3kb in bioassay and Plasmodium infection samples. (XLSX)

Acknowledgments

The authors thank the inhabitants of the collection sites for their support during the study.

Author Contributions

Conceptualization: Charles S. Wondji.

Data curation: Leon M. J. Mugenzi, Jack Hearn, Charles S. Wondji.

Formal analysis: Leon M. J. Mugenzi, Gareth D. Weedall, Jack Hearn, Charles S. Wondji.

Funding acquisition: Charles S. Wondji.

Investigation: Theofelix A. Tekoh, Stevia T. Ntadoun, Achille D. Chi, Mahamat Gadji, Benjamin D. Menze, Magellan Tchouakui.

Methodology: Leon M. J. Mugenzi, Charles S. Wondji.

Project administration: Charles S. Wondji.

Resources: Helen Irving, Murielle J. Wondji.

Visualization: Leon M. J. Mugenzi, Jack Hearn, Charles S. Wondji.

Writing - original draft: Leon M. J. Mugenzi, Jack Hearn, Charles S. Wondji.

Writing – review & editing: Leon M. J. Mugenzi, Gareth D. Weedall, Jack Hearn, Charles S. Wondji.

References

- 1. WHO. World malaria report 2023. Report No.: 978-92-4-008617-3.
- Bhatt S, Weiss DJ, Cameron E, Bisanzio D, Mappin B, Dalrymple U, et al. The effect of malaria control on Plasmodium falciparum in Africa between 2000 and 2015. Nature. 2015; 526(7572):207–11. https:// doi.org/10.1038/nature15535 PMID: 26375008
- Mosha JF, Kulkarni MA, Lukole E, Matowo NS, Pitt C, Messenger LA, et al. Effectiveness and costeffectiveness against malaria of three types of dual-active-ingredient long-lasting insecticidal nets (LLINs) compared with pyrethroid-only LLINs in Tanzania: a four-arm, cluster-randomised trial. Lancet. 2022; 399(10331):1227–41. https://doi.org/10.1016/S0140-6736(21)02499-5 PMID: 35339225
- Riveron JM, Huijben S, Tchapga W, Tchouakui M, Wondji MJ, Tchoupo M, et al. Escalation of Pyrethroid Resistance in the Malaria Vector Anopheles funestus Induces a Loss of Efficacy of Piperonyl Butoxide-Based Insecticide-Treated Nets in Mozambique. J Infect Dis. 2019; 220(3):467–75. https:// doi.org/10.1093/infdis/jiz139 PMID: 30923819
- Protopopoff N, Mosha JF, Lukole E, Charlwood JD, Wright A, Mwalimu CD, et al. Effectiveness of a long-lasting piperonyl butoxide-treated insecticidal net and indoor residual spray interventions, separately and together, against malaria transmitted by pyrethroid-resistant mosquitoes: a cluster, randomised controlled, two-by-two factorial design trial. Lancet. 2018; 391(10130):1577–88. https://doi.org/10.1016/S0140-6736(18)30427-6 PMID: 29655496
- Martinez-Torres D, Chandre F, Williamson M, Darriet F, Berge JB, Devonshire AL, et al. Molecular characterization of pyrethroid knockdown resistance (kdr) in the major malaria vector Anopheles gambiae ss. Insect Mol Biol. 1998; 7(2):179–84.
- Balabanidou V, Kampouraki A, MacLean M, Blomquist GJ, Tittiger C, Juárez MP, et al. Cytochrome P450 associated with insecticide resistance catalyzes cuticular hydrocarbon production in Anopheles gambiae. Proc Natl Acad Sci USA. 2016; 113(33):9268–73. https://doi.org/10.1073/pnas.1608295113 PMID: 27439866

- Kreppel K, Viana M, Main B, Johnson P, Govella N, Lee Y, et al. Emergence of behavioural avoidance strategies of malaria vectors in areas of high LLIN coverage in Tanzania. Sci Rep. 2020; 10(1):14527. https://doi.org/10.1038/s41598-020-71187-4 PMID: 32883976
- Challenger JD, Nash RK, Ngufor C, Sanou A, Toé KH, Moore S, et al. Assessing the variability in experimental hut trials evaluating insecticide-treated nets against malaria vectors. Curr Res Parasitol Vector Borne Dis. 2023; 3:100115. https://doi.org/10.1016/j.crpvbd.2023.100115 PMID: 36895438
- Irving H, Wondji CS. Investigating knockdown resistance (kdr) mechanism against pyrethroids/DDT in the malaria vector Anopheles funestus across Africa. BMC Genet. 2017; 18:1–11.
- Duneau D, Sun H, Revah J, San Miguel K, Kunerth HD, Caldas IV, et al. Signatures of insecticide selection in the genome of Drosophila melanogaster. G3: Genes, Genomes, Genetics. 2018; 8(11):3469–80.
- Weetman D, Wilding CS, Neafsey DE, Müller P, Ochomo E, Isaacs AT, et al. Candidate-gene based GWAS identifies reproducible DNA markers for metabolic pyrethroid resistance from standing genetic variation in East African Anopheles gambiae. Sci Rep. 2018; 8(1):2920. <u>https://doi.org/10.1038/</u> s41598-018-21265-5 PMID: 29440767
- Lucas ER, Miles A, Harding NJ, Clarkson CS, Lawniczak MK, Kwiatkowski DP, et al. Whole-genome sequencing reveals high complexity of copy number variation at insecticide resistance loci in malaria mosquitoes. Genome Res. 2019; 29(8):1250–61. <u>https://doi.org/10.1101/gr.245795.118</u> PMID: 31345938
- Alkan C, Coe BP, Eichler EE. Genome structural variation discovery and genotyping. Nat Rev Genet. 2011; 12(5):363–76. https://doi.org/10.1038/nrg2958 PMID: 21358748
- Alonge M, Wang X, Benoit M, Soyk S, Pereira L, Zhang L, et al. Major impacts of widespread structural variation on gene expression and crop improvement in tomato. Cell. 2020; 182(1):145–61. e23. https:// doi.org/10.1016/j.cell.2020.05.021 PMID: 32553272
- Brookfield JF. Evolutionary genetics: mobile DNAs as sources of adaptive change? Curr Biol. 2004; 14 (9):R344–R5. https://doi.org/10.1016/j.cub.2004.04.021 PMID: 15120086
- Weischenfeldt J, Symmons O, Spitz F, Korbel JO. Phenotypic impact of genomic structural variation: insights from and for human disease. Nature Reviews Genetics. 2013; 14(2):125–38. https://doi.org/10. 1038/nrg3373 PMID: 23329113
- Weedall GD, Mugenzi LMJ, Menze BD, Tchouakui M, Ibrahim SS, Amvongo-Adjia N, et al. A cytochrome P450 allele confers pyrethroid resistance on a major African malaria vector, reducing insecticide-treated bednet efficacy. Sci Transl Med. 2019; 11(484).
- Mugenzi LM, Menze BD, Tchouakui M, Wondji MJ, Irving H, Tchoupo M, et al. A 6.5-kb intergenic structural variation enhances P450-mediated resistance to pyrethroids in malaria vectors lowering bed net efficacy. Mol Ecol. 2020; 29(22):4395–411. https://doi.org/10.1111/mec.15645 PMID: 32974960
- Weedall GD, Riveron JM, Hearn J, Irving H, Kamdem C, Fouet C, et al. An Africa-wide genomic evolution of insecticide resistance in the malaria vector Anopheles funestus involves selective sweeps, copy number variations, gene conversion and transposons. PLoS Genet. 2020; 16(6):e1008822. https://doi. org/10.1371/journal.pgen.1008822 PMID: 32497040
- Barnes KG, Irving H, Chiumia M, Mzilahowa T, Coleman M, Hemingway J, Wondji CS. Restriction to gene flow is associated with changes in the molecular basis of pyrethroid resistance in the malaria vector Anopheles funestus. Proc Natl Acad Sci U S A. 2017; 114(2):286–91. https://doi.org/10.1073/pnas. 1615458114 PMID: 28003461
- 22. Mugenzi LMJ, Menze BD, Tchouakui M, Wondji MJ, Irving H, Tchoupo M, et al. Cis-regulatory CYP6P9b P450 variants associated with loss of insecticide-treated bed net efficacy against Anopheles funestus. Nat Commun. 2019; 10(1):4652.
- Wondji CS, Irving H, Morgan J, Lobo NF, Collins FH, Hunt RH, et al. Two duplicated P450 genes are associated with pyrethroid resistance in Anopheles funestus, a major malaria vector. Genome Res. 2009; 19(3):452–9. https://doi.org/10.1101/gr.087916.108 PMID: 19196725
- Tchouakui M, Fossog BT, Ngannang BV, Djonabaye D, Tchapga W, Njiokou F, Wondji CS. Investigation of the influence of a glutathione S-transferase metabolic resistance to pyrethroids/DDT on mating competitiveness in males of the African malaria vector, Anopheles funestus. Wellcome Open Res. 2019; 4:13. https://doi.org/10.12688/wellcomeopenres.15013.2 PMID: 31069259
- 25. Riveron JM, Osae M, Egyir-Yawson A, Irving H, Ibrahim SS, Wondji CS. Multiple insecticide resistance in the major malaria vector Anopheles funestus in southern Ghana: implications for malaria control. Parasites & vectors. 2016; 9:1–9. https://doi.org/10.1186/s13071-016-1787-8 PMID: 27628765
- 26. Mugenzi LM, Akosah-Brempong G, Tchouakui M, Menze BD, Tekoh TA, Tchoupo M, et al. Escalating pyrethroid resistance in two major malaria vectors Anopheles funestus and Anopheles gambiae (sl) in Atatam, Southern Ghana. BMC Infect Dis. 2022; 22(1):799.

- Koekemoer L, Kamau L, Hunt R, Coetzee M. A cocktail polymerase chain reaction assay to identify members of the Anopheles funestus (Diptera: Culicidae) group. American Journal of Tropical Medicine and Hygiene. 2002; 66(6):804–11. https://doi.org/10.4269/ajtmh.2002.66.804 PMID: 12224596
- WHO. Standard operating procedure for testing insecticide susceptibility of adult mosquitoes in WHO tube tests2022.
- WHO. Guidelines for laboratory and field-testing of long-lasting insecticidal nets. World Health Organization; 2013. Report No.: 9241505273.
- Ghurye J, Koren S, Small ST, Redmond S, Howell P, Phillippy AM, Besansky NJ. A chromosome-scale assembly of the major African malaria vector Anopheles funestus. Gigascience. 2019; 8(6):giz063. https://doi.org/10.1093/gigascience/giz063 PMID: 31157884
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The sequence alignment/map format and SAMtools. Bioinformatics. 2009; 25(16):2078–9. https://doi.org/10.1093/bioinformatics/btp352 PMID: 19505943
- 32. Broad-institute. Picard 2019 [Available from: https://broadinstitute.github.io/picard/.
- Rajaby R, Sung W-K. TranSurVeyor: an improved database-free algorithm for finding non-reference transpositions in high-throughput sequencing data. Nucleic Acids Res. 2018; 46(20):e122-e. https://doi.org/10.1093/nar/gky685 PMID: <u>30137425</u>
- Kohany O, Gentles AJ, Hankus L, Jurka J. Annotation, submission and screening of repetitive elements in Repbase: RepbaseSubmitter and Censor. BMC Bioinformatics. 2006; 7(1):1–7. https://doi.org/10. 1186/1471-2105-7-474 PMID: 17064419
- Huynh N, Depner N, Larson R, King-Jones K. A versatile toolkit for CRISPR-Cas13-based RNA manipulation in Drosophila. Genome Biology. 2020; 21(1). https://doi.org/10.1186/s13059-020-02193-y PMID: 33203452
- Ferretti L, Ramos-Onsins SE, Pérez-Enciso M. Population genomics from pool sequencing. Mol Ecol. 2013; 22(22):5561–76. https://doi.org/10.1111/mec.12522 PMID: 24102736
- Gel B, Serra E. karyoploteR: an R/Bioconductor package to plot customizable genomes displaying arbitrary data. Bioinformatics. 2017; 33(19):3088–90. https://doi.org/10.1093/bioinformatics/btx346 PMID: 28575171
- Hall TA, editor BioEdit: a user-friendly biological sequence alignment editor and analysis program for Windows 95/98/NT. Nucleic Acids Symp Ser; 1999: Oxford.
- 39. Riveron JM, Irving H, Ndula M, Barnes KG, Ibrahim SS, Paine MJ, Wondji CS. Directionally selected cytochrome P450 alleles are driving the spread of pyrethroid resistance in the major malaria vector Anopheles funestus. Proc Natl Acad Sci U S A. 2013; 110(1):252–7. <u>https://doi.org/10.1073/pnas.1216705110 PMID: 23248325</u>
- Schmittgen TD, Livak KJ. Analyzing real-time PCR data by the comparative CT method. Nat Protoc. 2008; 3(6):1101–8. https://doi.org/10.1038/nprot.2008.73 PMID: 18546601
- Gearing LJ, Cumming HE, Chapman R, Finkel AM, Woodhouse IB, Luu K, et al. CiiiDER: A tool for predicting and analysing transcription factor binding sites. PLoS One. 2019; 14(9):e0215495. <u>https://doi.org/10.1371/journal.pone.0215495</u> PMID: 31483836
- 42. Mathelier A, Fornes O, Arenillas DJ, Chen C-y, Denay G, Lee J, et al. JASPAR 2016: a major expansion and update of the open-access database of transcription factor binding profiles. Nucleic Acids Res. 2016; 44(D1):D110–D5. https://doi.org/10.1093/nar/gkv1176 PMID: 26531826
- 43. Kalsi M, Palli SR. Transcription factors, CncC and Maf, regulate expression of CYP6BQ genes responsible for deltamethrin resistance in Tribolium castaneum. Insect Biochem Mol Biol. 2015; 65:47–56. https://doi.org/10.1016/j.ibmb.2015.08.002 PMID: 26255690
- Amezian D, Nauen R, Le Goff G. Transcriptional regulation of xenobiotic detoxification genes in insects —An overview. Pestic Biochem Physiol. 2021; 174:104822. <u>https://doi.org/10.1016/j.pestbp.2021.</u> 104822 PMID: 33838715
- 45. Nkemngo FN, Mugenzi LM, Tchouakui M, Nguiffo-Nguete D, Wondji MJ, Mbakam B, et al. Xeno-monitoring of molecular drivers of artemisinin and partner drug resistance in P. falciparum populations in malaria vectors across Cameroon. Gene. 2022; 821:146339. <u>https://doi.org/10.1016/j.gene.2022</u>. 146339 PMID: 35183684
- Bass C, Nikou D, Blagborough AM, Vontas J, Sinden RE, Williamson MS, Field LM. PCR-based detection of Plasmodium in Anopheles mosquitoes: a comparison of a new high-throughput assay with existing methods. Malar J. 2008; 7:1–9.
- Snounou G. Genotyping of Plasmodium spp. In: Doolan DL, editor. Malaria Methods and Protocols: Methods and Protocols. Totowa, NJ: Humana Press; 2002. p. 103–16.
- Metcalfe CJ, Casane D. Modular organization and reticulate evolution of the ORF1 of Jockey superfamily transposable elements. Mobile DNA. 2014; 5:1–13.

- Nefedova LN, Kuzmin IV, Makhnovskii PA, Kim AI. Domesticated retroviral GAG gene in Drosophila: New functions for an old gene. Virology. 2014;450–451:196–204.
- Nefedova L, Gigin A, Kim A. Domesticated LTR-Retrotransposon gag-Related Gene (Gagr) as a Member of the Stress Response Network in Drosophila. Life (Basel). 2022; 12(3). https://doi.org/10.3390/ life12030364 PMID: 35330115
- Njoroge H, Van't Hof A, Oruni A, Pipini D, Nagi SC, Lynd A, et al. Identification of a rapidly-spreading triple mutant for high-level metabolic insecticide resistance in Anopheles gambiae provides a real-time molecular diagnostic for antimalarial intervention deployment. Mol Ecol. 2022; 31(16):4307–18. https://doi.org/10.1111/mec.16591 PMID: 35775282
- Chen S, Li X. Transposable elements are enriched within or in close proximity to xenobiotic-metabolizing cytochrome P450 genes. BMC Evol Biol. 2007; 7(1):1–13. <u>https://doi.org/10.1186/1471-2148-7-46</u> PMID: <u>17381843</u>
- Catania F, Kauer M, Daborn P, Yen J, Ffrench-Constant R, Schlötterer C. World-wide survey of an Accord insertion and its association with DDT resistance in Drosophila melanogaster. Mol Ecol. 2004; 13(8):2491–504. https://doi.org/10.1111/j.1365-294X.2004.02263.x PMID: 15245421
- Guio L, Barrón MG, González J. The transposable element Bari-Jheh mediates oxidative stress response in Drosophila. Mol Ecol. 2014; 23(8):2020–30. https://doi.org/10.1111/mec.12711 PMID: 24629106
- Daborn P, Boundy S, Yen J, Pittendrigh B, Ffrench-Constant R. DDT resistance in Drosophila correlates with Cyp6g1 over-expression and confers cross-resistance to the neonicotinoid imidacloprid. Mol Genet Genomics. 2001; 266:556–63. https://doi.org/10.1007/s004380100531 PMID: 11810226
- 56. Chung H, Bogwitz MR, McCart C, Andrianopoulos A, Ffrench-Constant RH, Batterham P, Daborn PJ. Cis-regulatory elements in the Accord retrotransposon result in tissue-specific expression of the Drosophila melanogaster insecticide resistance gene Cyp6g1. Genetics. 2007; 175(3):1071–7. <u>https://doi.org/10.1534/genetics.106.066597</u> PMID: 17179088
- Fukaya T, Levine M. Transvection. Current Biology. 2017; 27(19):R1047–R9. <u>https://doi.org/10.1016/j.</u> cub.2017.08.001 PMID: 29017034
- Veith A, Moorthy B. Role of cytochrome p450s in the generation and metabolism of reactive oxygen species. Curr Opin Toxicol. 2018; 7:44–51. <u>https://doi.org/10.1016/j.cotox.2017.10.003</u> PMID: 29527583
- Molina-Cruz A, Dejong RJ, Charles B, Gupta L, Kumar S, Jaramillo-Gutierrez G, Barillas-Mury C. Reactive Oxygen Species Modulate Anopheles gambiae Immunity against Bacteria and Plasmodium. Journal of Biological Chemistry. 2008; 283(6):3217–23. https://doi.org/10.1074/jbc.M705873200 PMID: 18065421
- Félix RC, Müller P, Ribeiro V, Ranson H, Silveira H. Plasmodium infection alters Anopheles gambiae detoxification gene expression. BMC Genomics. 2010; 11(1):1–10. https://doi.org/10.1186/1471-2164-11-312 PMID: 20482856
- David J-P, Strode C, Vontas J, Nikou D, Vaughan A, Pignatelli PM, et al. The Anopheles gambiae detoxification chip: a highly specific microarray to study metabolic-based insecticide resistance in malaria vectors. Proc Natl Acad Sci U S A. 2005; 102(11):4080–4. <u>https://doi.org/10.1073/pnas.</u> 0409348102 PMID: 15753317
- Müller P, Donnelly MJ, Ranson H. Transcription profiling of a recently colonised pyrethroid resistant Anopheles gambiae strain from Ghana. BMC Genomics. 2007; 8(1):1–12. https://doi.org/10.1186/ 1471-2164-8-36 PMID: 17261191
- Tchouakui M, Chiang MC, Ndo C, Kuicheu CK, Amvongo-Adjia N, Wondji MJ, et al. A marker of glutathione S-transferase-mediated resistance to insecticides is associated with higher Plasmodium infection in the African malaria vector Anopheles funestus. Sci Rep. 2019; 9(1):5772. https://doi.org/10.1038/ s41598-019-42015-1 PMID: 30962458
- Ndiath MO, Cailleau A, Diedhiou SM, Gaye A, Boudin C, Richard V, Trape J-F. Effects of the kdr resistance mutation on the susceptibility of wild Anopheles gambiae populations to Plasmodium falciparum: a hindrance for vector control. Malar J. 2014; 13(1):340. <u>https://doi.org/10.1186/1475-2875-13-340</u> PMID: 25176292